Author's response to reviews

Title: Global Challenges with Scale-Up of the Integrated Management of Childhood Illness Strategy: Results of a Multi-country Survey

Authors:

Ameena E Goga (Ameena.Goga@mrc.ac.za)
Lulu M Muhe (muhel@who.int)

Version: 5 Date: 27 June 2011

Author's response to reviews:

The authors list has been re-written in the suggested style.
The authors affiliations have been reviewed and the format has been revised.
The abstract has been reviewed and is similar to the main abstract in the body of the manuscript.
The authors contributions has been reviewed. All authors read and approved the final manuscript.
Tables have been moved to the body of the main manuscript; shading has been removed and vertical text has been converted to horizontal text.
The Figures are an integral part of the manuscript so they remain as Figures. I am sorry but I am not sure what was meant by columns and tabulated. So apart from changing the numbers of the figures the figures remain the same.
The references have been reviewed and et.al. has been replaced by authors names
Highlighting has been removed.
The manuscript has been read again and typos removed.